Author's response to reviews

Title: Association between Asymptomatic Hyperuricemia and New-onset Chronic Kidney Disease in Japanese Male Workers: A Long-term Retrospective Cohort Study

Authors:

Masatoshi Kawashima (m-kawashima@umin.ac.jp)
Koji Wada (kwada-sgy@umin.ac.jp)
Hiroshi Ohta (ohta.hiroshi.md@gmail.com)
Hiroyuki Terawaki (terawaki@iikei.ac.jp)
Yoshiharu Aizawa (aizawa@kitasato-u.ac.jp)

Version: 4 Date: 19 June 2011

Author's response to reviews: see over
Reply for Editor's report comment

MS: 1111533793463033
Title: Association between Asymptomatic Hyperuricemia and New-onset Chronic Kidney Disease in Japanese Male Workers: A Long-term Retrospective Cohort Study
Masatoshi Kawashima, Koji Wada, Hiroshi Ohta, Hiroyuki Terawaki and Yoshiharu Aizawa

We wish to express my gratitude that you accepted my manuscript for publication. We send revised manuscript that changed as your instruction major revisions and minor revisions.